Author’s response to reviews

Title: Oscillometric Blood Pressure By Age And Height For Non Overweight Children And Adolescents In Lubumbashi, Democratic Republic of Congo

Authors:

Emmanuel Muyumba (elmuyumba_01@yahoo.fr)
Dophra Nkulu (ngoynkuludophra@gmail.com)
Clarence Mukeng (deckeng@gmail.com)
Jacques Musung (jamusung@yahoo.fr)
Placide Kakoma (pkambola@yahoo.fr)
Christian Kakisingi (chriskakis@yahoo.fr)
Justin Kizonde (kizondej@hotmail.com)
Françoise Malonga (kaj_malonga@hotmail.com)
Oscar Luboya (oscarluboya@yahoo.fr)
Olivier Mukuku (oliviermukuku@yahoo.fr)
Weili Yan (yanwl@fudan.edu.cn)

Version: 6 Date: 25 Jun 2017

Author’s response to reviews:

Dear Editor

We would like to thank you for your comments.

We have remplace this sentence: “The approval to publish the study was obtained from a parent or guardian of the child or youth.” by this one: “The informed written consent for the publication of individual clinical details was obtained from the parent/guardians of the child or adolescent.”